5 July 2021

Editorial Office

World Journal of Clinical Cases

Dear Editors

We are grateful for your acceptance of our manuscript. We confirm that our manuscript is original, is not under consideration by other journals, and that the material has not been published elsewhere. All authors approved the final version of the manuscript and agree to be accountable for all aspects of this work. The authors have no conflicts of interest to declare.

We answer your suggestion of our manuscript.

(1) The title is too long, and it should be no more than 18 words;

We shortened our manuscript title. Please check it.

(2) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor;

We documented powerpoint which contains original images which is used in figures.

(3) PMID and DOI numbers are missing in the reference list. Please provide the PubMed numbers and DOI citation numbers to the reference list and list all authors of the
We corrected our manuscript. We write all writers and PMID and DOI. However some articles we used for reference do not have PMID and DOI. Please check it and if all references need it, please inform us.

Yours sincerely,

Motomura Kenta, M.D., Ph.D.

Department of Hepatology, IIZUKA HOSPITAL, 3-83 Yoshio-machi, Iizuka, Fukuoka 820-8505, Japan.

Tel: +81-948-22-3800 Fax: +81-948-29-5744

E-mail: kmotomurah2@aih-net.com